

Citation: Roberts HE, Hurst J, Robinson N, Brown H, Flanagan P, Vass L, et al. (2015) Structured Observations Reveal Slow HIV-1 CTL Escape. PLoS Genet 11(2): e1004914. doi:10.1371/journal. pgen.1004914

Editor: Takashi Gojobori, National Institute of Genetics, JAPAN

Received: July 25, 2014

Accepted: November 23, 2014

Published: February 2, 2015

Copyright: © 2015 Roberts et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data from the SPARTAC study cannot be made publically available due to ethical reasons. The data are available for interested researchers, who meet the criteria for access, through the Medical Research Council Clinical Trials Unit, UK.

Funding: The SPARTAC trial was funded by a grant from the Wellcome Trust (069598/Z/02/Z). JF is supported by the Medical Research Council and the National Institute for Health Research Oxford Biomedical Research Centre. REP, ARM, JH, JF are supported by the Oxford Martin School. HER is supported by the Medical Research Council and RESEARCH ARTICLE

Structured Observations Reveal Slow HIV-1 CTL Escape

Hannah E. Roberts^{1‡}, Jacob Hurst^{1,2‡}, Nicola Robinson^{1,3}, Helen Brown^{1,3}, Peter Flanagan¹, Laura Vass¹, Sarah Fidler⁴, Jonathan Weber⁴, Abdel Babiker⁵, Rodney E. Phillips^{1,2,3}*, Angela R. McLean^{2,6‡}, John Frater^{1,2,3‡}, SPARTAC trial investigators¹¹

 The Peter Medawar Building for Pathogen Research, Nuffield Department of Clinical Medicine, Oxford University, Oxford, United Kingdom, 2 The Institute for Emerging Infections, The Oxford Martin School, Oxford, Oxford United Kingdom, 3 Oxford NIHR Comprehensive Biomedical Research Centre, Oxford, United Kingdom, 4 Division of Medicine, Wright Fleming Institute, Imperial College, London, United Kingdom, 5 Medical Research Council Clinical Trials Unit, London, United Kingdom, 6 Department of Zoology, Oxford University, Oxford, United Kingdom

‡ HER and JH contributed equally to this work. ARM and JF also contributed equally to this work.
¶ Membership of the SPARTAC Trial Investigators is listed in the Acknowledgments.

rodney.phillips@ndm.ox.ac.uk

Abstract

The existence of viral variants that escape from the selection pressures imposed by cytotoxic T-lymphocytes (CTLs) in HIV-1 infection is well documented, but it is unclear when they arise, with reported measures of the time to escape in individuals ranging from days to years. A study of participants enrolled in the SPARTAC (Short Pulse Anti-Retroviral Therapy at HIV Seroconversion) clinical trial allowed direct observation of the evolution of CTL escape variants in 125 adults with primary HIV-1 infection observed for up to three years. Patient HLA-type, longitudinal CD8+ T-cell responses measured by IFN-y ELISpot and longitudinal HIV-1 gag, pol, and nef sequence data were used to study the timing and prevalence of CTL escape in the participants whilst untreated. Results showed that sequence variation within CTL epitopes at the first time point (within six months of the estimated date of seroconversion) was consistent with most mutations being transmitted in the infecting viral strain rather than with escape arising within the first few weeks of infection. Escape arose throughout the first three years of infection, but slowly and steadily. Approximately one third of patients did not drive any new escape in an HLA-restricted epitope in just under two years. Patients driving several escape mutations during these two years were rare and the median and modal numbers of new escape events in each patient were one and zero respectively. Survival analysis of time to escape found that possession of a protective HLA type significantly reduced time to first escape in a patient (p = 0.01), and epitopes escaped faster in the face of a measurable CD8+ ELISpot response (p = 0.001). However, even in an HLA matched host who mounted a measurable, specific, CD8+ response the average time before the targeted epitope evolved an escape mutation was longer than two years.

Nuffield Department of Medicine. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Author Summary

The cytotoxic T-lymphocyte (CTL) arm of the immune response is thought to play a significant role in the control of HIV-1 infection. Mutations within the HIV-1 genome allow the virus to escape recognition by CTLs and so evade the immune response. These escape mutations have been well documented but observed waiting times to escape within an individual have ranged from days to years. Many studies describing CTL escape have taken a detailed look at a few patients. Our analysis is based on a cohort of 125 clinical trial participants with immunologic and viral sequence data taken at regular longitudinal time points within the first few years of infection. Results suggested that the majority of CTL-related mutations present early in infection had been transmitted in the infecting viral strain as opposed to arising in the new host due to selection pressure imposed by CTLs. Whilst the prevalence of CTL escape mutations in the dataset was high, the incidence of new escape was relatively low; around one third of patients did not drive an escape within the first two years. Patients possessing a 'protective' HLA genotype had a significantly shorter waiting time to first escape than those without.

Introduction

The HIV-1-specific cytotoxic T-lymphocyte (CTL) response begins as early as 2 to 3 weeks after infection [1], and there is evidence to suggest it may play an important role in the early control of viraemia [2, 3]. The onset of the response coincides with the decline of viral load. Certain host HLA genotypes (and therefore certain specific responses) have been found to be significantly associated with delayed progression to AIDS [4–6]. In addition to this, some studies based on the depletion of CD8⁺ T-cells in SIV-infected macaques have shown that the CTL response makes a crucial contribution to viral control in this animal model of AIDS [7]. However, others have questioned the role of CTL in controlling productive infection, both in the SIV-macaque model and in humans [8, 9].

Many studies have demonstrated the capability of the HIV-1-specific CTL response to select for viral variants that escape recognition by $CD8^+$ T-cells or prevent antigen presentation by HLA class I molecules and so evade the immune response [10–14]. Indeed, work done to quantify the effects of the various selective forces acting on HIV has found that 53% of non-*env* mutations that rise to fixation in the first few years represent viral adaptation to $CD8^+$ T-cell responses [15] whilst a detailed deep sequencing study of one patient found that the majority of early mutations were CTL related [16].

CTL escape has been observed to occur throughout the different stages of HIV-1 infection, from soon after seroconversion [13, 17-20] to many years into chronic infection [11, 21, 22]. Some epitopes consistently escape earlier and more frequently than others [23-26] but time to escape also varies considerably between patients and is likely to be influenced by factors such as strength and quality of the CD8⁺ T-cell response as well as pre-existence of compensatory mutations in sequences surrounding epitopes [11, 21, 27]. Escape pathways are complex; in many cases low frequency variants arise early, replacing the transmitted sequence, but it is not until some time later that one escape mutant, often containing multiple amino acid mutations, emerges and rises to fixation [2, 28-30]. In the absence of selective pressure from CTLs, reversion of escape mutations back to wild type has been documented [20, 31].

In this study we analysed the timing and extent of CTL escape in 125 patients across the HIV-1 *gag*, *pol*, *nef* and *env* genes. We used longitudinal viral sequence data covering the first three years of infection to count escape events in each of 46 epitopes in all available patient

samples, and only considered periods when patients were off therapy. Approximately one third of patients drove escape in an epitope for which they were HLA-matched within their first year of infection and escape continued to be seen into the second year. However an estimated 33% of patients had still not driven any escape in their HLA-restricted epitopes across *gag*, *pol* or *nef* by the end of their second year off therapy. In looking at whether escape was clustered in patients, we found little evidence to suggest that this was the case. We also observed that time to escape for individual epitopes was faster in HLA-matched patients and in patients who were able to mount a measurable CD8⁺ T-cell response, and that patients with protective HLA alleles were more likely to have their first escape sooner.

Results

Cohort characteristics

The cohort consisted of 125 treatment-naive, HIV-1 subtype B infected adults who were recruited soon after the estimated date of seroconversion (median 11 weeks; range 1–20 weeks) as part of the SPARTAC clinical trial and followed for up to three years (see <u>Methods</u>, full details in [<u>32</u>], patient characteristics shown in <u>Table 1</u>). Participants were randomised into one of three arms, which determined whether they received no treatment, 12 weeks antiretroviral therapy (ART) or 48 weeks ART. Consensus viral sequences for four immunogenic genes (*gag*, *pol*, *env* and *nef*), and IFN- γ CD8⁺ T-cell ELISpot data were collected longitudinally. Participants were followed for up to three years whilst off ART (median 109 weeks, IQR 85–145 weeks); some participants initiated long-term ART before the end of the three years, upon

Table 1. Patient characteristics table.

Characteristic	Whole trial (N = 366)	UK patients (N = 125)
Sex—no. (%)		
Male	219 (60)	125 (100)
Female	147 (40)	0 (0)
Age—yr		
Median (interquartile range)	32 (25–40)	34 (29–42)
Range	19–63	19–60
Estimated time since seroconversion at randomisation—wk		
Median (interquartile range)	12 (9–15)	11 (7–14)
Range	1–24	1–20
CD4+ count— cells/mm ³		
Median (interquartile range)	559 (435–700)	550 (435–682)
Range	95–1399	130–1310
HIV RNA level—copies/ml		
Median (interquartile range)	4.53 (3.67–5.18)	4.78 (4.08-5.29)
Range	1.40-6.47	2.22-6.22

For this study, we took all the UK HIV-1 Subtype B infected participants that had been enrolled in the SPARTAC trial and for which exact times of initiation and cessation of both short course ART and, if applicable, long term ART were known. The resulting cohort of 125 participants had similar characteristics to the whole trial cohort except that all the UK HIV-1B infected participants recruited were male. Participants in the UK cohort were also recruited slightly earlier in infection on average, and (perhaps consequently) had slightly higher viral loads on average at recruitment compared to the trial as a whole.

doi:10.1371/journal.pgen.1004914.t001

reaching the trial clinical endpoint. The amount of data available at each time point is summarised in <u>S1 Table</u>.

Escape prevalence at baseline and after 1 year

A CTL escape mutation was defined as *any amino acid variation in previously identified (and phenotypically confirmed) escape sites within optimal defined epitopes* in *gag, pol, env* or *nef* (see <u>Methods</u> for more details). All analyses were also run using alternative, more lenient definitions of escape. Those additional analyses, which are shown in <u>S7–S12</u> Figs., confirm that the results presented here are not dependent on this definition of escape. In addition to drawing on a large body of collated literature to form this definition we also validated it by analysing the strength of patients' CD8⁺ T-cell responses to autologous epitope variants (see <u>Methods</u> and <u>S2 Table</u>).

At baseline (the first trial study visit) the data comprised a total of 3831 epitope sequences from 122 patients in which we were able to look for variation within known escape sites. The range of times from seroconversion amongst these patients matched that of the whole cohort. Of these, 1235 contained escape mutations. However, this escape at baseline was, for almost all epitopes, equally prevalent in HLA-matched and HLA-mismatched hosts (Fig. 1A). This pattern is more consistent with variation that has been transmitted than variation that has been driven by within-host HLA-restricted immunity prior to the first sampling. In contrast, after one year, four epitopes derived from *gag*, *pol* and *nef* had markedly higher escape prevalence in HLA-matched hosts (Fig. 1B). Further analysis (see Methods for details) confirmed that the distribution of points in Fig. 1B (at one year) is significantly different



Figure 1. Escape prevalence in HLA-matched and-mismatched hosts. Epitopes are labelled where possible with their first three amino acids, and length if necessary to distinguish them. Bars represent 95% binomial confidence limits (Agresti-Coull method). The dotted black line is y = x. Epitopes labelled in bold have a significantly higher proportion of escape in matched hosts than in mismatched hosts (one-tailed Fisher's Exact test, p < 0.1, note that multiple testing was not corrected for since the purpose of these tests was simply to give a relative measure of significance). Raw data for the number of patients in each category can be found in S3 and S4 Tables. First visit (baseline) sequence data was available in at least one gene for 122/125 patients. The range of time from seroconversion among these patients was 1–20 weeks, median 11. (A) At baseline the majority of epitopes have escape at equal prevalence in HLA-matched and-mismatched hosts (B) 52 weeks later within-host evolution has resulted in a higher prevalence of escape in HLA-matched hosts for many epitopes, and in the distribution as a whole HLA-matching and escape are now significantly associated (p < 0.01, permutation test, see <u>Methods</u> for details).

doi:10.1371/journal.pgen.1004914.g001

from what would be expected under the null hypothesis that escape is randomly distributed between matched and mismatched hosts, where as the distribution at baseline is not.

Incidence of escape during up to three years of observation

Of those epitopes sequenced, 2297 were initially wild-type and had at least one later sequence for comparison, and hence were candidates to allow the observation of incident escape (note that in a few cases this initial time point was after baseline due to sequence data being unavailable). Due to the high variability of the *env* gene, the vast majority of epitopes in *env* (all 32 HLA-matched patient-epitope pairs and 215/316 HLA-mismatched patient-epitope pairs) were already escaped at baseline, and hence whilst these were included in our analyses we were not able to comprehensively study incident escape in env. In the 2297 candidate epitopes, just 108 'escape' events were observed in patients studied while off ART and only 37 of those were in epitopes in an HLA-matched patient. To see if there was evidence for a separate group of particularly fast escaping patients we compared the distribution of the number of escape events per host (in their HLA-restricted epitopes) with a random distribution, calculated using a Poisson process, and found no significant difference between the distributions (Fig. 2A). In fact this Poisson process in which escape was driven at an average rate of 0.0018 per week across all HLA-matched patient-epitope pairs, described the distribution of incident escape across patients well. Since many patients had missing data, particularly at later time points (S1 Table) we used this average rate to construct a similar expected distribution supposing all patients had had a full set of sequence data for the first two years (whilst untreated) (Fig. 2B). Even after accounting for the missing data in this way, only 12% of patients would have driven more than two escape variants in HLA-restricted epitopes and the median and modal numbers of incident escape events were one and zero respectively.

This analysis is meaningful if patients with missing data would have had similar rates of escape to those observed for a full two years whilst untreated. We believe this to be a valid assumption and assess it in further detail later.

When does escape happen?

Time to escape can be described either for each patient in a population (how long does it take until a viral variant with a new escape mutation is detectable in a patient?) or for each epitope within each patient (how long does it take until an escape mutation appears in an initially wild-type epitope?). By considering the first such escape event (in patients or in epitopes) the well-developed statistical tools of survival analysis become applicable. Since only 9% of patients have more than one incident escape event in an epitope for which they are HLA-matched (Fig. 2A) very little information is lost in such analysis.

In assessing the suitability of these analyses we also needed to address the question of whether the time at which patients are censored from the survival analysis (which occurs when they have no further sequence data) is independent of the rate at which they drive escape mutations. For some patients the fact that we were unable to obtain sequence data at later time points was a direct consequence of the fact that they had reached the trial primary endpoint (a CD4 cell count < 350 cells /µ l or commencing long-term ART) within the time frame of our observation period. Hence if, as some might hypothesise, escape events can drive clinical failure, it is possible that censored patients would have high rates of escape that we'd fail to observe.

We looked in more detail at those patients who reached the trial endpoint (or clinically 'failed') within our three-year observation period. An incident escape was observed in 6/20 patients for whom we had no sequence data after clinical failure compared to 7/24 patients who did have sequence data after clinical failure. In the 7 patients in the latter group who drove at



Figure 2. The distribution of incident escape across patients. (A) The observed distribution of incident escape (in HLA-matched patient-epitope pairs) across patients (red line) is similar to the distribution given by a Poisson process where the average rate of escape is constant across all patient-epitope pairs (black line). (B) The expected distribution of escape supposing all patients in (A) had a full set of sequence data and were observed for a full 2 years whilst off-ART. A goodness-of-fit test showed that there was no significant difference between the observed and expected distributions in (A) (p > 0.1). Details of calculations for the expected distributions are given in the Methods.

doi:10.1371/journal.pgen.1004914.g002

least one escape event, there were just two cases where an escape mutation was first seen at the time point after clinical failure (see <u>S1 Text</u> and <u>S2 Fig.</u> for more details). Therefore the extent to which escape is linked to censoring time is small, and it is fair to assume that the (unobserved) rate of escape in patients who are censored from the analysis is the same as that of those who continue to be observed.

Fig. 3 shows time to first escape for patients whilst they were off ART. Only patients with at least two time points of sequence data for both *nef* and *gag* (the genes for which the most data were available) were used to avoid underestimating the occurrence of escape due to missing data. This left 65 patients for this analysis. Note that this number includes both treated and untreated patients (range of previous ART duration 2–12 months). We also considered these patient groups separately (S4–S6 Figs.) and this did not change our conclusions. Multivariate Cox models for time to escape confirmed that the impact of early limited treatment on CTL escape was negligible (S1 Text).

The Kaplan-Meier plot for time to first escape in an HLA-restricted epitope shows that there is large variation between patients in time to first escape, with some patients who received: 12 weeks of therapy escaping before the 24 week time point (so that the midpoint



Figure 3. Kaplan-Meier plots of time to first escape using a midpoint approximation. Patients who were either missing data for *gag* or *nef*, the two genes for which the most data was available, were not included so as not to skew the results due to lack of data. 4 of these patients had no epitopes restricted by their HLA types that were WT at baseline, so n = 61 initially here. (A) Time to first escape in an HLA-restricted epitope (solid line) is plotted along with the 95% confidence intervals (dotted lines). (B) Patients are split according to whether they have one of the more 'protective' HLA alleles or not. The set of beneficial alleles was taken to be B*58, B*27, B*57, A*26, B*51, A*11, B*14, B*18, B*08 (all the HLA-A and-B alleles down to B*08 that are present in our data, taken from the ranking in [47]) as this split the patients approximately in half. Having a protective HLA resulted in a significantly increased risk of HLA-matched escape (p = 0.01, Likelihood Ratio test on Cox Proportional Hazards model with single predictor. Hazard ratio = 3.7, 95% C.I. = (1.2, 11.3)). In both plots the x-axis represents the time since cessation of treatment, or baseline for those not receiving treatment, and vertical checks mark time points at which patients were censored (either because they began long term ART or because there was no further sequence data available for them). Numbers indicate the number of patients who have not yet escaped or been censored at the corresponding time points marked by red dots (0, 27, 55, 93 wks).

doi:10.1371/journal.pgen.1004914.g003

approximation gives the therapy-adjusted time to escape as around 6 weeks, assuming no viral evolution on ART) whilst an estimated 33% of patients did not drive any new escape in their HLA-restricted epitopes in 92 weeks off ART (95% C.I. (16,70)%). Note also that the slope of the curve is close to constant; there is no indication of a propensity for many incident escape events to arise early during the observation period and fewer later.

Having a protective HLA type is associated with earlier first escape

Although there was no evidence for a separate group of patients with a high frequency of escape (Fig. 2), there was evidence that some patients had a tendency to drive escape mutations sooner than others. Fig. 3B shows a Kaplan-Meier plot for time to first HLA-matched escape event for patients split according to whether or not they have one of the more protective HLA types. First escape is earlier in patients with a 'protective' HLA allele than in those without.

Waiting time to escape for a wild-type epitope

<u>Fig. 4</u> records Kaplan-Meier survival curves for wild-type epitopes. This analysis included all patient epitope pairs which were observed whilst the patient was off ART (i.e. in which it would have been possible to observe escape), giving a total of 46 epitopes and 114 patients. Here 'failure' is the appearance of an escape mutation in an epitope. The characteristic time to escape for a wild-type epitope is of the order of months or years. It is faster in HLA-matched hosts (Fig. 4A), and amongst epitopes in HLA-matched hosts it is faster if there is a detectable

 $CD8^+$ T-cell ELISpot response to that epitope in that host (Fig. 4B). Yet even in a host with the epitope's restricting HLA and a detectable $CD8^+$ T-cell response more than half of wild-type epitopes still show no escape after two years' observation (Fig. 4B, survival = 0.69 between 84 and 115 weeks).

Discussion

The literature on CTL escape in HIV-1 infection is dominated by studies of small numbers of individuals [17, 18, 33–35]. Several publications describe single patients in whom many CTL escape mutations arise and recent papers have described very active, rapid outgrowth of many escape mutations in a small number of patients studied intensively in the first weeks of infection [2, 19]. It has been known for some time that escape mutations that arise in early infection have a faster fixation rate than those that arise during chronic infection [10, 36]. These observations lead to a natural set of questions about the timing of CTL escape in HIV-1 infection. What is a normal number of escape events observable in a patient during the first few years of infection? Is escape very concentrated amongst a subset of patients so that patients fall into two groups—one with lots of escape and one with little? Are escape events also focussed in time so that much escape occurs early in infection and very little later on?

To address questions like this properly it is necessary to follow a large number of patients through time, so that those in whom nothing happens are given proper attention. This study of 125 patients in the first few years of untreated infection from a large randomised controlled trial does just that. It finds that the incidence of escape is very low; after two years of observation an estimated 33% of patients had not seen a first escape incident in *gag*, *pol* or *nef*. The observed frequency of incident escape across all epitopes in HLA-matched hosts translated to an estimated



Figure 4. Kaplan-Meier survival curves for wild-type epitopes. For the 114 patients and 46 epitopes for which appropriate sequence data was available: (A) Time to escape in epitopes within HLA-mismatched (dark blue) and-matched (light blue) hosts is shown. (B) Epitopes in HLA-matched hosts are split according to whether they have a measurable CD8⁺ T-cell ELISpot response to them (light red) or not (dark red). Escape is strongly significantly faster in epitopes within HLA-matched hosts ($p < 1 \times 10^{-4}$), hazard ratio = 2.6, 95% C.I. = (1.7, 4.2)) and then faster again if the host has a measurable CD8⁺ T-cell ELISpot response to that epitope (p = 0.013, hazard ratio = 3.0, 95% C.I. = (1.3, 7.0)). Note that p-values and hazard ratios were calculated using a mixed effects model with random intercept terms for both patient and epitope ID, to account for correlation of observations within patients and within epitopes (see Methods).

doi:10.1371/journal.pgen.1004914.g004

12% of patients driving more than 2 escape mutations within 2 years (Fig. 2B). The inevitable conclusion is that patients with many escape events are unusual and, further, there was no evidence for a separate group of patients with a very high number of escape events (Fig. 2A). Equally, the distribution of escape through time showed no evidence for much escape early on and little later; the Kaplan Meier survival curves for the proportion of patients without an escape, and the proportion of epitopes without escape both have constant slopes (Figs. <u>3</u> & <u>4</u>).

It has long been known that some HLA types confer a survival advantage upon their host. The relationship between the restricting HLA type and the propensity for an epitope to escape has been contentious. We previously [37] analysed the prevalence of escape in 84 chronically infected patients and concluded that protective T-cell responses were associated with more frequent viral escape, hypothesising that this result was due to a high fitness cost of mutations in epitopes restricted by protective alleles. However, Asquith [38], in a study that involved measuring the entropy of epitopes across sequences in the LANL HIV database, found that epitope entropy was significantly positively correlated with relative hazard of the restricting HLA allele, concluding that protective HLA alleles restricted epitopes that escape less frequently than average. In this cohort patients with protective HLA alleles escape significantly sooner (Fig. 3B), and epitopes restricted by protective HLAs escape faster (although not significantly so) than others (S3 Fig.). Further, all patients with escape in two or more HLA restricted epitopes had at least one of our defined list of protective HLAs. This large cohort study thus supports our previous finding that protective HLAs are associated with more frequent escape.

At first glance it seems counterintuitive that patients with some HLA types escape sooner and more frequently than others but that there was no evidence for a separate group of patients with much escape. However this apparent conflict is easily resolved; it is likely that patients with protective HLAs drive escape more frequently but do not have so much escape as to form a distinct subset of patients. They just lie at the top end of the distribution.

Why is escape in this cohort so slow? Is it because we missed very early escape events that occurred before patients were recruited? Is it that escape is concentrated in a small number of epitopes in which the process of escape is already largely complete? Or is it that escape is rare because responses are rare? Finally, is the definition of escape used here too restrictive?

Liu et al [19] have documented the presence of minor epitope variants within days of seroconversion, rising to fixation quickly in the first few weeks. However our results, presented in Fig. 1A, suggest that there was little escape happening this early on. Since reversion of escape mutations in mismatched hosts is relatively slow [23] we used the proportion of escape in HLA-mismatched hosts at first time point as an estimate of the prevalence of transmitted escape for each epitope, and hence by comparison with the proportion of escape in HLAmatched hosts obtained a quantitative estimate for the frequency of escape before the first time point. As an example, for TST (an epitope that is known to escape particularly early [39]), at the first time point 45% of HLA-mismatched patients had an escaped form of the epitope. We therefore assume that 45% of the 7 HLA-matched patients would have been infected with an escaped epitope. Hence, with reference to S5 Table, the estimated number of escape events in TST that have arisen since baseline is $6-7 \times 0.45 = 2.9$. Summing over all epitopes, these calculations gave the estimated 'survival' of wild-type epitopes in the weeks between seroconversion and first available sequence data (an average of 18.5 weeks) as 98.9% compared to 97.1% in the first 12 weeks of observation in which patients were off ART (see <u>S1 Text</u> for further details). Hence, whilst we have missed a handful of escape events (: 6 in total), these data do not support the idea that the majority of escape happens very early, within the first few weeks or months of infection.

The high prevalence of escape at baseline might suggest that the incidence of escape is low because many of the epitopes that are prone to acquire escape mutations already have them

transmitted at baseline. This was true of the *env* epitopes we studied, which is unsurprising given the various and strong selection pressured on *env*, and a few others such as Gag TST. However, whilst there were epitopes in which the process of escape was complete even at baseline, this was not the general rule; Fig. 4 shows that more than 70% of HLA-matched patient-epitope pairs remain unescaped throughout the three-year observation period. Nor is it the case that all of the fastest escaping epitopes end up escaping in all HLA-matched patients. In these data only two epitopes (KYK and TAF) escape in all HLA-matched patients within the observation period.

The fastest rate of escape was seen within epitopes against which a positive CD8⁺ T-cell response had been measured (a group comprising: 30% of HLA-matched host-epitope pairs). Even here the incidence of escape was low (Fig. 4B). Hence we cannot conclude that it was a lack of strong CD8⁺ T-cell responses in the hosts to the set of epitopes we studied that led to few escape events being observed. However, as only IFN- γ production was measured to indicate the strength of the response we cannot tell whether or not the responses measured were qualitatively good in other ways.

Lastly, the definition of escape that we have used may have prevented some true escape from being classified as such. We were able expand the definition in two ways: by considering amino acid mutations that occur outside of known escape sites (but still within the epitope), and by extending the list of epitopes beyond the set of optimal defined epitopes (see <u>S1 Text</u> and <u>S6 Table</u>). Repeated analyses given these less stringent definitions are presented in <u>S7–S12</u> Figs. Both of these amendments reduce the time to first escape in a patient so that an estimated: 10% (down from: 30%) of patients have driven no escape in HLA-restricted epitopes within the first two years. Changing the definition does not change the conclusion that the majority of escape at baseline is transmitted, or that escape mutations rise to fixation with a roughly uniform frequency across the observation period. However, extending the list of epitopes diminishes the association between HLA relative hazard and time to escape, suggesting that the strength of this association is sensitive to the set of epitopes considered.

We acknowledge that even this extended epitope list is unlikely to be complete, and that the Los Alamos 'A-list' will evolve to include new peptides and HLA-restrictions [40, 41]. Indeed, when using the extended list we observe 144 vs. 69 incident 'escapes' in HLA-mismatched vs.-matched patient-epitope pairs (S12 Fig.). Many of these 'escapes' in supposedly unrestricted epitopes cannot be accounted for by overlap with another, HLA-restricted epitope. Some of these may be indicative of immune escape driven by other components of the immune response such as NK cells, CD4⁺ T-cells or the antibody response. But understanding the impact of these alternative selective pressures is not necessary for assessing the concern that our methods are blind to CTL escape occurring in previously unidentified epitopes. We have previously shown that HLA-association studies favour the detection of rapid escape mutations. Hence any escape in unknown or poorly characterised epitopes that we have not looked in is likely to be slower than the time frame of this study and consequently not affect our conclusions [42].

Our definition of escape was also based on consensus viral sequence data; hence we were unable to detect the presence of any low frequency variants unlike with deep sequencing methods [2, 17]. However, variants that never rise to a majority in the population are of limited interest since they lack biological significance in terms of being unlikely to effect viral dynamics or the course of the disease in the long term. Variants that do rise to greater than 50% prevalence would be detectable with our methods. Therefore our use of consensus viral sequence is not problematic.

The final aspect of our definition of escape to address is the use of HXB2 as our comparison sequence. Although there is no 'gold standard' comparator for this analysis, there were only

four epitopes in which HXB2 differed from the 'A list' optimal at a documented escape site. We therefore re-ran the analysis re-classifying these four epitopes and found no significant difference in the distribution of baseline escape prevalence between HLA-matched and-mismatched hosts. We did identify 4 new incident escapes (3 in EVK, 1 in RPN). Although the EVK escape variant is only poorly defined, its inclusion would mean that an estimated 25% of patients had not driven an escape by the end of 2 years off-therapy (as opposed to 33%).

In conclusion, this study of a cohort from a randomised controlled trial, the proper design for assessing rates of escape across a population, reveals that escape is on average both less frequent and less concentrated in the first few months of infection than would be expected from recent intense studies of much smaller numbers of acutely infected individuals.

Materials and Methods

Ethics statement

The SPARTAC trial was approved by the following authorities: Medicines and Healthcare products Regulatory Agency (UK), Ministry of Health (Brazil), Irish Medicines Board (Ireland), Medicines Control Council (South Africa), and The Uganda National Council for Science and Technology (Uganda). It was also approved by the following ethics committees in the participating countries: Central London Research Ethics Committee (UK), Hospital Universitrio Clementino Fraga Filho Ethics in Research Committee (Brazil), Clinical Research and Ethics Committee of Hospital Clinic in the province of Barcelona, Spain, The Adelaide and Meath Hospital Research Ethics Committee (Ireland), University of Witwatersrand Human Research Ethics Committee, University of Kwazulu-Natal Research Ethics Committee and University of Cape Town Research Ethics Committee (South Africa), Uganda Virus Research Institute Science and ethics committee (Uganda), The Prince Charles Hospital Human Research Ethics Committee and St Vincent's Hospital Human Research Ethics Committee (Australia), and the National Institute for Infectious Diseases Lazzaro Spallanzani, Institute Hospital and the Medical Research Ethics Committee, and the ethical committee Of the Central Foundation of San Raffaele, MonteTabor (Italy). All participants signed a written informed consent.

Participants and trial design

The design of the SPARTAC trial is reported elsewhere [32]. In brief, SPARTAC was an international open Randomised Controlled Trial enrolling adults with PHI within 6 months of a last negative, equivocal or incident HIV-1 test. All participants gave written informed consent. The trial was approved by research ethics committees in each country. Time of seroconversion was estimated as the midpoint of last negative/equivocal and first positive tests, or date of incident test. Participants were randomised to receive ART for 48 weeks (ART-48), 12 weeks (ART-12) or no therapy (standard of care, SOC). The primary endpoint was a composite of two events: if participants either reached a CD4 count of <350 cells/mm³ (> 3 months after randomisation and confirmed within 4 weeks) or initiated long-term ART. This provided an immunological surrogate of clinical progression, but also allowed inclusion of those participants who commenced ART at CD4 cell counts greater than cells/mm³.

The participants studied here consisted of a sub-group 125 HIV-1 subtype B positive individuals of whom 42, 41 and 42 were randomised to ART-12, ART-48 and SOC, respectively. Peripheral blood mononuclear cells (PBMCs) were collected at regular intervals during the treatment period and follow-up. The sub-group of 125 participants was based on those individuals infected with subtype B HIV-1 for whom both sequence and ELISpot data were available, predominantly determined by specimen availability.

Sequences

Viral RNA was extracted from patient plasma (Qiagen Viral RNA extraction kit) and the HIV-1 *pol, gag, nef* and *env* genes were amplified separately using nested PCR reactions and primers described previously [20, 37, 43]. The PCR products were purified and sequenced using Big Dye dideoxy terminator chemistry (ABI). Sequences are therefore presented as the consensus from a bulk PCR product for each time-point. Sequence data were available for each patient at a subset of the following time points: 0, 16 or 24, 52 or 60, 108 and 156 weeks following recruitment onto the trial (see <u>S1 Table</u> for details).

CD8 T cell ELISpot assays

Quantification of CD8⁺ T-cell response was performed by interferon gamma ELISpot analysis in all subtype B patients, using methods described elsewhere [44]. Responses were determined to 181 subtype B optimal peptides covering the Gag (n = 67), Pol (n = 47), Nef (n = 30) and Env (n = 37) proteins and 195 autologous variants synthesised followed sequence analysis. The optimal peptides tested were derived from the "best-defined 'A list' CTL epitope" database accessed from the Los Alamos HIV Immunology Website. Analyses were performed singly, but in duplicate for the negative control, and results expressed as spot forming units (SFU) per 10⁶ cells. ELISpot responses were measured at 0, 24 and 60 weeks post-recruitment.

HLA genotyping

Patients' HLA type was determined to the oligo-allelic level using Dynal RELITM Reverse Sequence-Specific Oligonucleotide kits for the HLA-A, -B and -C loci (Dynal Biotech). To obtain four-digit typing, Dynal Biotech Sequence-Specific priming kits were used, in conjunction with the Sequence-Specific Oligonucleotide type.

Defining escape

A CTL escape mutation was defined as *any amino acid variation in previously identified (and phenotypically confirmed) escape sites within an 'optimal' defined epitope.* Known escape sites were defined as within-epitope sites for which escape (E), calculated escape (CE), inferred escape (IE) or literature escape (LE) had been documented in the 'CTL/CD8⁺ Epitope Variants and Escape Mutations' HIV database table, as downloaded on 05/11/12 [45]. Any amino acid difference between autologous sequence and HXB2 in one of the sites under consideration was classed as escape. The list of 'optimal' defined epitopes was initially drawn from the set of epitopes for which ELISpot assays had been conducted (see above) though this was later extended (S1 Text and S6 Table). The 4-digit HLA-type restricting each epitope was determined using the 'Best-defined' epitope summary table on the Los Alamos HIV Immunology Database [45] downloaded on 01/03/2013. 2-digit HLA types were used only where 4-digit HLA types were not available. Only epitopes restricted by HLA-A and-B alleles were included since data on patient HLA-C alleles was not initially available. However a subsequent analysis found there were no incident escape mutations observed in any of these patients in epitopes restricted by their HLA-C alleles, so the exclusion of these epitopes is not problematic.

This intentionally stringent definition of escape was chosen to optimise the balance between including all relevant variation available in sequence data whilst excluding variation that is not escape. This definition does not include escape mutations that lie outside the epitope and disrupt proteasomal processing or antigen presentation pathways. It also excludes escape that occurs in poorly studied or unknown epitopes [46]. To complement the published data available and to check that mutations classed as 'escape' genuinely did result in diminished IFN- γ

production by CD8⁺ T-cells, a comparison was made between the magnitudes of CD8⁺ T-cell ELISpot responses for the pre-defined optimal peptides with those for autologous epitope variants observed in the population at baseline. Variants for 7 Gag and 2 Pol epitopes were tested. In all cases except Gag EVD (where the optimal peptide differed from the HXB2 sequence) the major variant(s) elicited a diminished response. In 6/9 cases responses to all 'escape' variants were reduced by at least 90% (Table S2).

Identification of escape in the data

Firstly, the sequence data was subjected to quality control: A phylogenetic tree was constructed for all sequences of each gene in turn so that patients whose sequences did not cluster with each other could be identified and removed from the dataset if necessary. Due to difficulties in aligning all regions of all genes, epitopes were located using a string-matching search, allowing for 4 amino acid differences between the HXB2 epitope sequence and the patient sequence. Spurious matches were identified by hand and removed. Missing data resulted in escape being classed as 'not determined' if the HXB2 and viral sequence matched at all of the defined escape sites in the epitope except those where data was missing.

Statistics

Permutation test on the distribution of escape between HLA-matched and-mismatched hosts. We wished to test the null hypothesis that escape is equally likely to be present in HLA-matched and-mismatched hosts against the alternative hypothesis that there is more escape in HLA-matched hosts. The statistic used for this was the mean (across all epitopes) of the Fisher's exact test p-values for whether escape was independent of HLA-matching for that epitope. A permutation test was used to calculate the distribution of this mean under the null hypothesis; 2,000 random data sets were generated by permuting which patients the escape events occurred in for each epitope (i.e. there was no permutation between epitopes). The mean of the p-values for the observed data was then calculated and compared to the null distribution (S1 Fig.).

Poisson process for calculation of expected escape distributions. For each epitope in each patient in which an incident escape could have been observed, the total amount of time for which the patient was observed whilst off ART was calculated. Let E be the total number of escape events observed in the population, T be the total observation time (whilst off ART) across all epitopes in all patients and t_p be the total observation time for all epitopes of a particular patient, p. Then, under the assumption that the average rate of incidence of escape in all patients is r = E/T, the probability of patient p driving i escape events is:

$$\Pr(p \text{ drives } i \text{ escape events during observation}) = \frac{e^{-t_p r} (t_p r)^i}{i!}$$

The expected number of patients with i incident escape events is then the sum over all patients of these probabilities. The expected distribution of escape across different epitopes is calculated similarly.

To account for missing data (for Fig. 2B), it was supposed that each patient had 108 weeks of data, during which time they were off ART. Where an epitope sequence was entirely missing (so that whether or not the epitope was initially wild-type could not be determined), the probability of it being wild-type at baseline was taken to be equal to the prevalence of escape at the first time point, as in <u>S5 Table</u> (this was necessary for a total of 203 HLA-matched patient-epitope pairs). We could then calculate $t_p = (\#HLA-matched, initially wild-type epitopes) \times 108$.

Note that this model does not impose the condition that only one escape is possible in each patient-epitope pair. We believe this to be a reasonable simplification given that the total number of escape events is small.

A X^2 goodness-of-fit test with 4 degrees of freedom (i.e. categories for 0, 1, 2, 3, and > 4 escaped epitopes) was used to test the null hypothesis that the observed data followed the expected distribution.

Survival analysis. For the survival analysis shown in Figs. $\underline{3}$ and $\underline{4}$ only patient-epitope pairs in which it would have been possible to observe an escape were considered. This included all patient-epitope pairs that were wild-type in known escape sites at the first data point and had data available at a second data point. For each patient-epitope pair where an incident escape occurred, escape times were estimated as the midpoint between the last wild-type observation (or if that observation was during therapy, the therapy end time) and the first appearance of the escape mutant in the consensus sequence (or the clinical endpoint, whichever was earlier). Escape events occurring during treatment were discarded. Where no incident escape occurred, right-censoring started at the last sequence data time point or clinical endpoint, whichever was earlier. In Fig. 3, hazard ratios were calculated using a simple Cox Proportional Hazards model with a single predictor—the variable of interest. p-values relate to the Likelihood Ratio test comparing the one-predictor model with the null model. In Fig. 4, where observations from all patients and epitopes were pooled, hazard ratios and associated p-values were calculated using a mixed effects Cox model with a single predictor but also random intercept terms for patient and epitope ID. This was implemented with the package 'coxme' in R.

Supporting Information

S1 Text. Supplementary methods describing estimation of number of escape events before the first time point, analysis of whether censoring from the survival analysis is independent of escape, multivariate Cox models for time to escape and extending the definition of escape. (PDF)

S1 Fig. The distribution of escape between HLA-matched and mismatched host-epitope pairs at baseline and 1 year later. Histograms show the distribution of the mean of the Fisher's exact test p-values for independence of HLA-matching and escape in each epitope, under the null hypothesis that 'escape' is equally likely to be seen in HLA-matched and-mismatched hosts. (A) at baseline, (B) at 52 weeks. The red lines show the observed mean of the p-values. 2000 random simulations were used to obtain the null distributions. There is significant evidence to reject the null hypothesis at 52 weeks but not baseline, suggesting that little escape has already been driven before baseline. (EPS)

S2 Fig. Defining the relationship between clinical failure and censoring. Each patient-epitope pair was categorised according to if and when the patient reached the trial endpoint and how their time of clinical failure related to the last available sequence data point, as shown here diagrammatically. The three categories are (A) censoring may have resulted from clinical failure and there were no observations after failure, (B) censoring may have resulted from clinical failure but there was at least 1 further observation after failure, and (C) censoring is not related to clinical failure since long course ART was started after the last observed data point. Numbers on the right hand side represent the number of HLA-matched patient-epitope pairs and in brackets the number of distinct patients in each category. Note that three patients have epitopes in both A and B. One patient was removed as they were suspected to have been superinfected during the observation period. (EPS)

S3 Fig. Kaplan-Meier survival cure for wild-type epitopes, split by restricting HLA. All HLA-matched host-epitope pairs are considered. Epitopes restricted by a protective HLA-allele as defined in Fig. 3 (light red) escape sooner than others (dark red). There is no significant increased risk associated with restriction by a protective allele here (p = 0.3, mixed effects Cox model as in Fig. 4), though narrowing the definition of 'protective' does result in significance. (EPS)

S4 Fig. Kaplan-Meier plots to show time to first escape (using midpoint approximation) within the patients of this cohort, split by the duration of therapy they have received. Patients who were either missing data for *gag* or *nef*, the two genes for which the most data was available, were not included so as not to skew the results due to lack of data. Time to first escape in an HLA-restricted epitope is plotted. These survival curves are very similar for the patients who have received no ART and who have received < 6 months of ART. Patients who have received > 6 mo ART were observed for less time off ART and hence the number of patients in this group with enough data to be included in the analysis is smaller (15 patients compared to 22 with no ART and 24 with < 6 mo ART). Therefore, while there appears to be a slower rate of escape in patients who have received > 6 months ART this difference is not significant (p = 0.24) and the effect of removing this group from the data is small (see <u>S5 Fig.</u>). (EPS)

S5 Fig. Kaplan-Meier plots to show time to first escape (using midpoint approximation) within the patients of this cohort, excluding patients who have received > 6 mo ART. Patients who were either missing data for gag or nef, the two genes for which the most data was available, were not included so as not to skew the results due to lack of data. (A) Time to first escape in an HLA-restricted epitope (solid line) is plotted along with 95% confidence intervals (dotted line). (B) Patients are split according to whether they have one of the more 'protective' HLA alleles or not. The set of beneficial alleles was taken to be B*58, B*27, B*57, A*26, B*51, A*11, B*14, B*18, B*08 (all the HLA-A and-B alleles down to B*08 that are present in our data, taken from the ranking in [47]). Having a protective HLA resulted in a significantly increased risk of escape in an HLA-restricted epitope (p = 0.03, HR = 3.3). In both plots the x-axis represents the time since cessation of ART, or baseline for those who did not receive ART, and vertical checks mark time points at which patients were censored (either because they began long term ART or because there was no further sequence data available for them). Numbers indicate the number of patients who have not yet escaped or been censored at the corresponding red dots (time points: 0, 27, 55, 93 wks). (EPS)

S6 Fig. Kaplan-Meier survival curves for wild-type epitopes, excluding patients who have received > 6 mo ART. As with time to first escape in a patient, time to escape in epitopes was longer for epitopes in patients who have previously received > 6 mo ART, but there was no difference between epitopes in those who had received no ART and in those who had received < 6 mo ART. (A) Time to a first escape in epitopes within HLA-mismatched (dark blue) and-matched (light blue) hosts is shown. (B) Epitopes in HLA-matched hosts are split according to whether they have a measurable CD8⁺ T-cell ELISpot response to them (light red) or not (dark red). As in Fig. 4, escape is strongly significantly faster within epitopes in HLA-matched hosts ($p < 1 \times 10^{-4}$, HR = 2.5) and then faster again if the host has a measurable CD8⁺ T-cell ELISpot response to that epitope (p = 0.034, HR = 2.7). p-values and hazard ratios were calculated using a mixed effects model as in Fig. 4. (EPS)

S7 Fig. Escape prevalence in HLA-matched and-mismatched hosts, where escape is defined as any mutation within the epitope. Epitopes are labelled where possible with their first three amino acids. Bars represent 95% binomial confidence limits (Agresti-Coull method). The dotted black line is y = x. Epitopes labelled in bold have a significantly higher proportion of escape in matched hosts than in mismatched hosts (one-tailed Fisher's Exact test, p < 0.1). Note that this plot is noisier that Fig. 1 since a smaller proportion of the mutations classified as 'escape' are likely to be CTL related. Consequently the distribution at neither time point is significantly different from that under the null hypothesis that escape is equally likely to occur in HLA-matched and-mismatched hosts (permutation test, see Methods). (A) At baseline the majority of epitopes have escape at equal prevalence in HLAmatched and-mismatched hosts (B) 52 weeks later within host evolution has resulted in a higher prevalence of escape in HLA-matched hosts for many epitopes, with six lying significantly 'below the line' despite a decrease in the number of patients with sequence data available at 52 weeks.



S8 Fig. Kaplan-Meier plots to show time to first escape (using midpoint approximation) within the patients of this cohort, excluding patients who have received > 6 mo ART, where escape is defined as any mutation within the epitope. Patients who were either missing data for gag or nef, the two genes for which the most data was available, were not included so as not to skew the results due to lack of data. (A) Time to first escape in an HLA-restricted epitope (solid line) is plotted along with 95% confidence intervals (dotted line). (B) Patients are split according to whether they have one of the more 'protective' HLA alleles or not. The set of beneficial alleles was taken to be all the HLA-A and-B alleles down to B*08 that are present in our data, taken from the ranking in [47]. Having a protective HLA resulted in a significantly increased risk of escape in an HLA-restricted epitope (p = 0.003, HR = 3.6). In both plots the xaxis represents the time since cessation of ART, or baseline for those who did not receive ART, and vertical checks mark time points at which patients were censored (either because they began long term ART or because there was no further sequence data available for them). Numbers indicate the number of patients who have not yet escaped or been censored at the corresponding red dots (time points: 0, 27, 55 wks). (EPS)

S9 Fig. Kaplan-Meier survival curves for wild-type epitopes, excluding patients who have received > 6 mo ART, where escape is defined as any mutation within the epitope. (A) Time to a first escape for epitopes in HLA-mismatched (dark blue) and-matched (light blue) hosts is shown. (B) Epitopes within HLA-matched hosts are split according to whether they have a measurable CD8⁺ T-cell ELISpot response to them (light red) or not (dark red). As in Fig. 4, escape is strongly significantly faster within epitopes in HLA-matched hosts ($p < 1 \times 10^{-4}$, HR = 2.2) and then faster again if the host has a measurable CD8⁺ T-cell ELISpot response to that epitope (p = 0.001, HR = 2.9). p-values and hazard ratios were calculated using a mixed effects model as in Fig. 4.

(EPS)

S10 Fig. Escape prevalence in HLA-matched and-mismatched hosts, where the escape list has been extended beyond the optimals to all epitopes < 12 aa in length with a 4-digit HLA restriction defined. Epitopes are labelled where possible with their first three amino acids. Bars represent 95% binomial confidence limits (Agresti-Coull method). The dotted black line is y = x. Epitopes labelled in bold have a significantly higher proportion of escape in matched hosts than in mismatched hosts (one-tailed Fisher's Exact test, p < 0.1). (A) At baseline the

majority of epitopes have escape at equal prevalence in HLA-matched and-mismatched hosts and the distribution of points is not significantly different from as would be expected if escape was equally likely to occur in HLA-matched and-mismatched hosts (p = 0.5, permutation test) (B) 52 weeks later within host evolution has resulted in a higher prevalence of escape in HLAmatched hosts for many epitopes, with the eight epitopes and the distribution as a whole lying significantly 'below the line' (p < 0.003, permutation test, see <u>Methods</u>). (EPS)

S11 Fig. Kaplan-Meier plots to show time to first escape (using midpoint approximation) within the patients of this cohort, excluding patients who have received > 6 mo ART, where the escape list has been extended beyond the optimals to all epitopes < 12 as in length with a 4-digit HLA restriction defined. Patients who were either missing data for gag or nef, the two genes for which the most data was available, were not included so as not to skew the results due to lack of data. (A) Time to first escape in an HLA-restricted epitope (solid line) is plotted along with 95% confidence intervals (dotted line). (B) Patients are split according to whether they have one of the more 'protective' HLA alleles or not. The set of beneficial alleles was taken to be all the HLA-A and-B alleles down to B*08 that are present in our data, taken from the ranking in [47]. Having a protective HLA resulted in an increased risk of escape in an HLA-restricted epitope early on, but this was not significant. In both plots the x-axis represents the time since cessation of ART, or baseline for those who did not receive ART, and vertical checks mark time points at which patients were censored (either because they began long term ART or because there was no further sequence data available for them). Numbers indicate the number of patients who have not yet escaped or been censored at the corresponding time points marked by red dots 0, 27, 55 wks. (EPS)

S12 Fig. Kaplan-Meier survival curves for wild-type, epitopes excluding patients who have received > 6 mo ART, where the escape list has been extended beyond the optimals to all epitopes < 12 aa in length with a 4-digit HLA restriction defined. Time to a first escape for epitopes in HLA-mismatched (dark blue) and-matched (light blue) hosts is shown. As in Fig. 4, escape is strongly significantly faster within epitopes in HLA-matched hosts ($p < 1 \times 10^{-10}$, HR = 3.2). The effect of a response could not be analysed as CD8⁺ T-cell ELISpot responses were not measured for all of the extended list of epitopes. The p-value and hazard ratio was calculated using a mixed effects model as in Fig. 4. (EPS)

S1 Table. The number of individual patients for which viral sequence and CD8⁺ T-cell ELI-Spot data is available. Note that some sequences do not cover the whole gene and so the number given is an upper bound for the number of patients who have sequence data for any one epitope. For each patient ELISpot data was either available for all 181 optimal peptides across all four genes or not at all. Data was available in 115 patients at week 0, 109 patients at week 24 and 115 patients at week 60. The final column gives the number of patients with at least 2 time points of sequence data and ELISpot data at at least one of 0, 24 or 60 weeks. (PDF)

S2 Table. Diminished responses to variants present in the population at baseline. All epitopes for which CD8⁺ T-cell responses to autologous variants were assessed and for which there was at least one HLA-matched patient with a positive response to the optimal epitope are shown. The top sequence for each epitope gives the pre-defined optimal ELISpot peptide, which is also the most prevalent variant in each case. The 'Escape' column records whether or not each variant is classified as escape according to our definition. The

prevalence of each variant in the cohort at baseline and the proportion of patients with a particular variant who are HLA-matched for the epitope are also recorded. The final column gives the average ELISpot response to the variant (number of SFCs) in those that are HLA-matched and respond to the optimal, as a percentage of their response to the optimal.

(PDF)

S3 Table. The numbers of HLA-matched and-mismatched patients with escape at baseline. All p-values < 0.2 are given for a Fisher's exact test with the alternative hypothesis that the proportion of escape in matched hosts is greater than that in mismatched hosts. (PDF)

S4 Table. The numbers of HLA-matched and-mismatched patients with escape after 1 year. All p-values < 0.2 are given for a Fisher's exact test with the alternative hypothesis that the proportion of escape in matched hosts is greater than that in mismatched hosts. (PDF)

S5 Table. The numbers of HLA-matched and-mismatched patients with escape at the first time point. Since a few patients did not have available sequence data at baseline, this table is similar but not identical to <u>S3 Table</u>. The number of incident escape events observed within HLA-matched and-mismatched patients is also given. (PDF)

S6 Table. All epitopes in first extended definition of escape. Peptides that are also in the 'A-list' of optimal epitopes are shown in bold. (PDF)

Acknowledgments

We thank the participants of SPARTAC and the SPARTAC Trial Investigators: Trial Steering Committee: Independent Members- A Breckenridge(Chair), P Clayden, C Conlon, F Conradie, J Kaldor^{*}, F Maggiolo, F Ssali, Country Principal Investigators- DA Cooper, P Kaleebu, G Ramjee, M Schechter, G Tambussi, JM Miro, J Weber. Trial Physician: S Fidler. Trial Statistician: A Babiker. Data and Safety Monitoring Committee (DSMC): T Peto (Chair), A McLaren (in memoriam), V Beral, G Chene, J Hakim. Co-ordinating Trial Centre: Medical Research Council Clinical Trials Unit, London (A Babiker, K Porter, M Thomason, F Ewings, M Gabriel, D Johnson, K Thompson, A Cursley*, K Donegan*, E Fossey*, P Kelleher*, K Lee*, B Murphy*, D Nock*). Central Immunology Laboratories and Repositories: The Peter Medawar Building for Pathogen Research, University of Oxford, UK (R Phillips, J Frater, L Ohm Laursen*, N Robinson, P Goulder, H Brown). Central Virology Laboratories and Repositories: Jefferiss Trust Laboratories, Imperial College, London, UK (M McClure, D Bonsall*, O Erlwein*, A Helander*, S Kaye, M Robinson, L Cook*, G Adcock*, P Ahmed*). Clinical Endpoint Review Committee: N Paton, S Fidler. Investigators and Staff at Participating Sites: Australia: St Vincents Hospital, Sydney (A Kelleher), Northside Clinic, Melbourne (R Moore), East Sydney Doctors, Sydney (R McFarlane), Prahran Market Clinic, Melbourne (N Roth), Taylor Square Private Clinic, Sydney (R Finlayson), The Centre Clinic, Melbourne (B Kiem Tee), Sexual Health Centre, Melbourne (T Read), AIDS Medical Unit, Brisbane (M Kelly), Burwood Rd Practice, Sydney (N Doong), Holdsworth House Medical Practice, Sydney (M Bloch), Aids Research Initiative, Sydney (C Workman). Coordinating Centre in Australia: Kirby Institute University of New South Wales, Sydney (P Grey, DA Cooper, A Kelleher, M Law). Brazil: Projeto Praca Onze, Hospital Escola Sao Francisco de Assis, Universidade federal do Rio de Janeiro, Rio de Janeiro

(M Schechter, P Gama, M Mercon*, M Barbosa de Souza, C Beppu Yoshida, JR Grangeiro da Silva, A Sampaio Amaral, D Fernandes de Aguiar, M de Fatima Melo, R Quaresma Garrido). Italy: Ospedale San Raffaele, Milan (G Tambussi, S Nozza, M Pogliaghi, S Chiappetta, L Della Torre, E Gasparotto), Ospedale Lazzaro Spallanzani, Roma (G DOffizi, C Vlassi, A Corpolongo). South Africa: Cape Town: Desmond Tutu HIV Centre, Institute of Infectious Diseases, Cape Town (R Wood, J Pitt, C Orrell, F Cilliers, R Croxford, K Middelkoop, LG Bekker, C Heiberg, J Aploon, N Killa, E Fielder, T Buhler). Johannesburg: The Wits Reproductive Health and HIV Institute, University of Witswatersrand, Hillbrow Health Precinct, Johannesburg (H Rees, F Venter, T Palanee), Contract Laboratory Services, Johannesburg Hospital, Johannesburg (W Stevens, C Ingram, M Majam, M Papathanasopoulos). Kwazulu-Natal: HIV Prevention Unit, Medical Research Council, Durban (G Ramjee, S Gappoo, J Moodley, A Premrajh, L Zako). Uganda: Medical Research Council/Uganda Virus Research Institute, Entebbe (H Grosskurth, A Kamali, P Kaleebu, U Bahemuka, J Mugisha*, HF Njaj*). Spain: Hospital Clinic-IDIBAPS, University of Barcelona, Barcelona (JM Miro, M Lopez-Dieguez*, C Manzardo, JA Arnaiz, T Pumarola, M Plana, M Tuset, MC Ligero, MT Garca, T Gallart, JM Gatell). UK and Ireland: Royal Sussex County Hospital, Brighton (M Fisher, K Hobbs, N Perry, D Pao, D Maitland, L Heald), St Jamess Hospital, Dublin (F Mulcahy, G Courtney, S ODea, D Reidy), Regional Infectious Diseases Unit, Western General Hospital and Genitourinary Dept, Royal Infirmary of Edinburgh, Edinburgh (C Leen, G Scott, L Ellis, S Morris, P Simmonds), Chelsea and Westminster Hospital, London (B Gazzard, D Hawkins, C Higgs), Homerton Hospital, London (J Anderson, S Mguni), Mortimer Market Centre, London (I Williams, N De Esteban, P Pellegrino, A Arenas-Pinto, D Cornforth*, J Turner*), North Middlesex Hospital (J Ainsworth, A Waters), Royal Free Hospital, London (M Johnson, S Kinloch, A Carroll, P Byrne, Z Cuthbertson), Barts & the London NHS Trust, London (C Orkin, J Hand, C De Souza), St Marys Hospital, London (J Weber, S Fidler, E Hamlyn, E Thomson*, J Fox*, K Legg, S Mullaney*, A Winston, S Wilson, P Ambrose), Birmingham Heartlands Hospital, Birmingham (S Taylor, G Gilleran). Imperial College Trial Secretariat: S Keeling, A Becker. Imperial College DSMC Secretariat: C Boocock.

* Left the study team before the trial ended.

Author Contributions

Conceived and designed the experiments: REP JF SF JW AB. Performed the experiments: NR HB PF LV JF. Analyzed the data: HER ARM JH AB. Contributed reagents/materials/analysis tools: JH. Wrote the paper: HER JH REP AB ARM JF.

References

- McMichael AJ, Borrow P, Tomaras GD, Goonetilleke N, Haynes BF (2010) The immune response during acute HIV-1 infection: clues for vaccine development. Nat Rev Immunol 10: 11–23. doi: <u>10.1038/</u> <u>nri2674</u> PMID: <u>20010788</u>
- Goonetilleke N, Liu MK, Salazar-Gonzalez JF, Ferrari G, Giorgi E, et al. (2009) The first T cell response to transmitted/founder virus contributes to the control of acute viremia in HIV-1 infection. J Exp Med 206: 1253–1272. doi: 10.1084/jem.20090365 PMID: 19487423
- Kiepiela P, Ngumbela K, Thobakgale C, Ramduth D, Honeyborne I, et al. (2006) CD8+ T-cell responses to different HIV proteins have discordant associations with viral load. Nat Med 13: 46–53. doi: <u>10.1038/</u> <u>nm1520</u> PMID: <u>17173051</u>
- Pereyra F, Jia X, McLaren PJ, Telenti A, De Bakker P, et al. (2010) The major genetic determinants of HIV-1 control affect HLA class I peptide presentation. Science 330: 1551–1557. doi: <u>10.1126/science.</u> <u>1195271</u> PMID: <u>21051598</u>
- Fellay J, Shianna KV, Ge D, Colombo S, Ledergerber B, et al. (2007) A whole-genome association study of major determinants for host control of HIV-1. Science 317: 944–947. doi: <u>10.1126/science.</u> <u>1143767</u> PMID: <u>17641165</u>

- Altfeld M, Kalife ET, Qi Y, Streeck H, Lichterfeld M, et al. (2006) HLA Alleles Associated with Delayed Progression to AIDS Contribute Strongly to the Initial CD8+ T Cell Response against HIV-1. PLoS Med 3: e403+. doi: 10.1371/journal.pmed.0030403 PMID: 17076553
- Schmitz JE, Kuroda MJ, Santra S, Sasseville VG, Simon MA, et al. (1999) Control of viremia in simian immunodeficiency virus infection by CD8+ lymphocytes. Science 283: 857–860. doi: <u>10.1126/science.</u> 283.5403.857 PMID: 9933172
- Wong JK, Strain MC, Porrata R, Reay E, Sankaran-Walters S, et al. (2010) In vivo CD8+ T-cell suppression of SIV viremia is not mediated by CTL clearance of productively infected cells. PLoS Pathog 6: e1000748. doi: <u>10.1371/journal.ppat.1000748</u> PMID: <u>20126442</u>
- Kaiser P, Joos B, Niederöst B, Weber R, Günthard HF, et al. (2007) Productive human immunodeficiency virus type 1 infection in peripheral blood predominantly takes place in CD4/CD8 double-negative T lymphocytes. J Virol 81: 9693–9706. doi: <u>10.1128/JVI.00492-07</u> PMID: <u>17609262</u>
- Asquith B, Edwards CTT, Lipsitch M, McLean AR (2006) Inefficient Cytotoxic T Lymphocyte-Mediated Killing of HIV-1Infected Cells In Vivo. PLoS Biol 4: e90+. doi: <u>10.1371/journal.pbio.0040090</u> PMID: <u>16515366</u>
- Kelleher AD, Long C, Holmes EC, Allen RL, Wilson J, et al. (2001) Clustered mutations in HIV-1 gag are consistently required for escape from HLA-B27–restricted cytotoxic T lymphocyte responses. J Exp Med 193: 375–386. doi: <u>10.1084/jem.193.3.375</u> PMID: <u>11157057</u>
- Phillips RE, Rowland-Jones S, Nixon DF, Gotch FM, Edwards JP, et al. (1991) Human immunodeficiency virus genetic variation that can escape cytotoxic T cell recognition. Nature 354: 453–459. doi: <u>10.</u> <u>1038/354453a0</u> PMID: <u>1721107</u>
- Price DA, Goulder PJ, Klenerman P, Sewell AK, Easterbrook PJ, et al. (1997) Positive selection of HIV-1 cytotoxic T lymphocyte escape variants during primary infection. Proc Natl Acad Sci U S A 94: 1890–1895. doi: <u>10.1073/pnas.94.5.1890</u> PMID: <u>9050875</u>
- Zimbwa P, Milicic A, Frater J, Scriba TJ, Willis A, et al. (2007) Precise identification of a human immunodeficiency virus type 1 antigen processing mutant. J Virol 81: 2031–2038. doi: <u>10.1128/JVI.00968-06</u> PMID: 17108020
- Allen TM, Altfeld M, Geer SC, Kalife ET, Moore C, et al. (2005) Selective Escape from CD8+ T-Cell Responses Represents a Major Driving Force of Human Immunodeficiency Virus Type 1 (HIV-1) Sequence Diversity and Reveals Constraints on HIV-1 Evolution. J Virol 79: 13239–13249. doi: <u>10.1128/JVI.79.21.13239-13249.2005</u> PMID: <u>16227247</u>
- Henn MR, Boutwell CL, Charlebois P, Lennon NJ, Power KA, et al. (2012) Whole genome deep sequencing of HIV-1 reveals the impact of early minor variants upon immune recognition during acute infection. PLoS Pathog 8: e1002529. doi: <u>10.1371/journal.ppat.1002529</u> PMID: <u>22412369</u>
- Jones NA, Wei X, Flower DR, Wong M, Michor F, et al. (2004) Determinants of human immunodeficiency virus type 1 escape from the primary CD8+ cytotoxic T lymphocyte response. J Exp Med 200: 1243–1256. doi: <u>10.1084/jem.20040511</u> PMID: <u>15545352</u>
- Fischer W, Ganusov VV, Giorgi EE, Hraber PT, Keele BF, et al. (2010) Transmission of single HIV-1 genomes and dynamics of early immune escape revealed by ultra-deep sequencing. PloS One 5: e12303. doi: 10.1371/journal.pone.0012303 PMID: 20808830
- Liu MK, Hawkins N, Ritchie AJ, Ganusov VV, Whale V, et al. (2013) Vertical T cell immunodominance and epitope entropy determine HIV-1 escape. J Clin Invest 123: 380. doi: <u>10.1172/JCI65330</u> PMID: 23221345
- Duda A, Lee-Turner L, Fox J, Robinson N, Dustan S, et al. (2009) HLA-associated clinical progression correlates with epitope reversion rates in early human immunodeficiency virus infection. J Virol 83: 1228–1239. doi: 10.1128/JVI.01545-08 PMID: 19019964
- Jamieson BD, Yang OO, Hultin L, Hausner MA, Hultin P, et al. (2003) Epitope escape mutation and decay of human immunodeficiency virus type 1-specific CTL responses. J Immunol 171: 5372–5379. doi: 10.4049/jimmunol.171.10.5372 PMID: 14607940
- 22. Geels MJ, Cornelissen M, Schuitemaker H, Anderson K, Kwa D, et al. (2003) Identification of sequential viral escape mutants associated with altered T-cell responses in a human immunodeficiency virus type 1-infected individual. J Virol 77: 12430–12440. doi: <u>10.1128/JVI.77.23.12430-12440.2003</u> PMID: <u>14610167</u>
- Fryer HR, Frater J, Duda A, Roberts MG, Phillips RE, et al. (2010) Modelling the evolution and spread of HIV immune escape mutants. PLoS Pathog 6: e1001196. doi: <u>10.1371/journal.ppat.1001196</u> PMID: <u>21124991</u>
- Brumme ZL, Brumme CJ, Carlson J, Streeck H, John M, et al. (2008) Marked epitope-and allelespecific differences in rates of mutation in human immunodeficiency type 1 (HIV-1) Gag, Pol, and Nef cytotoxic T-lymphocyte epitopes in acute/early HIV-1 infection. J Virol 82: 9216–9227. doi: <u>10.1128/</u> JVI.01041-08 PMID: <u>18614631</u>

- Goulder PJ, Watkins DI (2004) HIV and SIV CTL escape: implications for vaccine design. Nat Rev Immunol 4: 630–640. doi: 10.1038/nri1417 PMID: 15286729
- 26. Klenerman P, Hill A (2005) T cells and viral persistence: lessons from diverse infections. Nat Immunol 6: 873–879. doi: 10.1038/ni1241 PMID: 16116467
- Chopera DR, Mlotshwa M, Woodman Z, Mlisana K, de Assis Rosa D, et al. (2011) Virological and immunological factors associated with HIV-1 differential disease progression in HLA-B* 58: 01-positive individuals. J Virol 85: 7070–7080. doi: 10.1128/JVI.02543-10 PMID: 21613398
- Herbeck JT, Rolland M, Liu Y, McLaughlin S, McNevin J, et al. (2011) Demographic processes affect HIV-1 evolution in primary infection before the onset of selective processes. J Virol 85: 7523–7534. doi: 10.1128/JVI.02697-10 PMID: 21593162
- 29. Oxenius A, Price DA, Trkola A, Edwards C, Gostick E, et al. (2004) Loss of viral control in early HIV-1 infection is temporally associated with sequential escape from CD8+ T cell responses and decrease in HIV-1-specific CD4+ and CD8+ T cell frequencies. J Infect Dis 190: 713–721. doi: <u>10.1086/422760</u> PMID: <u>15272399</u>
- Cao J, McNevin J, Holte S, Fink L, Corey L, et al. (2003) Comprehensive analysis of human immunodeficiency virus type 1 (HIV-1)-specific gamma interferon-secreting CD8+ T cells in primary HIV-1 infection. J Virol 77: 6867–6878. doi: 10.1128/JVI.77.12.6867-6878.2003 PMID: 12768006
- Leslie A, Pfafferott K, Chetty P, Draenert R, Addo M, et al. (2004) HIV evolution: CTL escape mutation and reversion after transmission. Nat Med 10: 282–289. doi: <u>10.1038/nm992</u> PMID: <u>14770175</u>
- Fidler S, Porter K, Ewings F, Frater J, Ramjee G, et al. (2013) Short-course antiretroviral therapy in primary HIV infection. N Engl J Med 368: 207–217. doi: <u>10.1056/NEJMoa1110039</u> PMID: <u>23323897</u>
- 33. Ferrari G, Korber B, Goonetilleke N, Liu MK, Turnbull EL, et al. (2011) Relationship between functional profile of HIV-1 specific CD8 T cells and epitope variability with the selection of escape mutants in acute HIV-1 infection. PLoS Pathog 7. doi: 10.1371/journal.ppat.1001273 PMID: 21347345
- Liu Y, McNevin J, Cao J, Zhao H, Genowati I, et al. (2006) Selection on the human immunodeficiency virus type 1 proteome following primary infection. J Virol 80: 9519–9529. doi: <u>10.1128/JVI.00575-06</u> PMID: <u>16973556</u>
- Salazar-Gonzalez JF, Salazar MG, Keele BF, Learn GH, Giorgi EE, et al. (2009) Genetic identity, biological phenotype, and evolutionary pathways of transmitted/founder viruses in acute and early HIV-1 infection. J Exp Med 206: 1273–1289. doi: 10.1084/jem.20090378 PMID: 19487424
- 36. Ganusov VV, Goonetilleke N, Liu MK, Ferrari G, Shaw GM, et al. (2011) Fitness costs and diversity of the cytotoxic T lymphocyte (CTL) response determine the rate of CTL escape during acute and chronic phases of HIV infection. J Virol 85: 10518–10528. doi: 10.1128/JVI.00655-11 PMID: 21835793
- Frater AJ, Brown H, Oxenius A, Günthard H, Hirschel B, et al. (2007) Effective T-cell responses select human immunodeficiency virus mutants and slow disease progression. J Virol 81: 6742–6751. doi: <u>10.</u> 1128/JVI.00022-07 PMID: 17409157
- Asquith B (2008) The Evolutionary Selective Advantage of HIV-1 Escape Variants and the Contribution of Escape to the HLA-Associated Risk of AIDS Progression. PLoS One 3: e3486. doi: <u>10.1371/journal.</u> pone.0003486 PMID: <u>18941529</u>
- Palmer D, Frater J, Phillips R, McLean AR, McVean G (2013) Integrating genealogical and dynamical modelling to infer escape and reversion rates in HIV epitopes. Proc Biol Sci 280. doi: <u>10.1098/rspb.</u> <u>2013.0696</u>
- Llano A, Williams A, Olvera A, Silva-Arrieta S, Brander C (2013) Best-Characterized HIV-1 CTL Epitopes: The 2013 Update. HIV Molecular Immunology 2013: 3–25.
- Frahm N, Yusim K, Suscovich TJ, Adams S, Sidney J, et al. (2007) Extensive HLA class I allele promiscuity among viral CTL epitopes. Eur J Immunol 37: 2419–2433. doi: <u>10.1002/eji.200737365</u> PMID: 17705138
- Fryer HR, Frater J, Duda A, Palmer D, Phillips RE, et al. (2012) Cytotoxic T-lymphocyte escape mutations identified by HLA association favor those which escape and revert rapidly. J Virol 86: 8568–8580. doi: 10.1128/JVI.07020-11 PMID: 22674992
- 43. English S, Katzourakis A, Bonsall D, Flanagan P, Duda A, et al. (2011) Phylogenetic analysis consistent with a clinical history of sexual transmission of HIV-1 from a single donor reveals transmission of highly distinct variants. Retrovirology 8: 54. doi: <u>10.1186/1742-4690-8-54</u> PMID: <u>21736738</u>
- 44. Frater J, Ewings F, Hurst J, Brown H, Robinson N, et al. (2014) HIV-1 specific CD4 responses in primary HIV-1 infection predict disease progression in the SPARTAC trial. AIDS.
- 45. Los Alamos National Laboratory (2013). HIV Molecular Immunology Database. URL <u>http://www.hiv.</u> lanl.gov/content/immunology.

- 46. Schellens IM, Kesmir C, Miedema F, van Baarle D, Borghans JA (2008) An unanticipated lack of consensus cytotoxic T lymphocyte epitopes in HIV-1 databases: the contribution of prediction programs. AIDS 22: 33–37. doi: <u>10.1097/QAD.0b013e3282f15622</u> PMID: <u>18090389</u>
- **47.** O'Brien SJ, Gao X, Carrington M (2001) HLA and AIDS: a cautionary tale. Trends Mol Med 7: 379–381. doi: <u>10.1016/S1471-4914(01)02131-1</u> PMID: <u>11530315</u>